Author’s response to reviews

Title: Exploring the Implementation of the Community for Successful Ageing (ComSA) program in Singapore: Lessons Learnt on Program Delivery for Improving BioPsychoSocial Health

Authors:

Su Aw (awsu@u.nus.edu)

Choon Huat Koh (ephkohch@nus.edu.sg)

Chuen Seng Tan (ephtcs@nus.edu.sg)

Mee Lian Wong (ephwml@nus.edu.sg)

Hubertus Vrijhoef (bert.vrijhoef@panaxea.eu)

Susana Harding (susanaharding@tsaofoundation.org)

Mary Ann Geronimo (mbgeronimo@tsaofoundation.org)

Zoe Hildon (zhildon1@jhu.edu)

Version: 2 Date: 03 Sep 2019

Author’s response to reviews:

1. We noticed authors Susana Harding and Mary Ann Geronimo are not included in the Authors’ contributions section. Please add their initials and individual contributions to this section.
   I have added that on pg 26 of the revised manuscript.

2. In accordance with our submission guidelines, please change the “Findings” heading to “Results”.
   Please see pg 12 of the revised manuscript.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
   Track changes and highlights have been removed.